Author's response to reviews

Title: Graduates from a reformed undergraduate medical curriculum based on Tomorrow's Doctors evaluate the effectiveness of their curriculum 6 years after graduation through interviews.

Authors:

Simon D Watmough (efcsw@liv.ac.uk)
Helen O'Sullivan (osulih@liv.ac.uk)
David CM Taylor (dcmt@liv.ac.uk)

Version: 3 Date: 29 September 2010

Author's response to reviews:

I have undertaken the revisions as suggested in the email from BMC dated 22/09/2010.

Hi

I have also split the first sentence in the first paragraph in "background" as I felt it was too long.

BW

Simon